Author’s response to reviews

Title: Mapping the interactome of overexpressed Raf kinase inhibitor protein in a gastric cancer cell line

Authors:

Huan Gu (dqngh@163.com)
Xianquan Zhan (yjzhan2011@gmail.com)
Guiying Zhang (guiyingzhang@hotmail.com)
Lu Yan (yanlu_152@163.com)
William CS Cho (williamcscho@gmail.com)
Maoyu Li (maoyuli@126.com)
Ting Liu (liuting818@126.com)
Zhuchu Chen (tcbl@xysm.net)

Version: 3  Date: 24 October 2013

Author’s response to reviews: see over
October 24, 2013

Editors of BMC Cancer

Dear Editors:

Many thanks for your decision to accept our manuscript, entitled “Mapping the interactome of overexpressed Raf kinase inhibitor protein in a gastric cancer cell line”. Here, we would like to resubmit the revised version changed by the editorial comments.

Enclosed are the point-by-point responses to the editorial comments.

We would like to hearing from you.

Sincerely,

Guiying Zhang, Ph.D., M.D.
Professor of Gastroenterology
Central South University Xiangya Hospital

Responses to the reviewer comments.

1. Author's Contribution:
For manuscripts with more than one author, all BMC Series journals require an Authors’ Contributions section to be placed after the Competing Interests section. An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. To qualify as an author one should 1) have made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; 2) have been involved in drafting the manuscript or revising it critically for important intellectual content; and 3) have given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content. Acquisition of funding, collection of data, or general supervision of the research group, alone, does not justify authorship. We suggest the following format (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES
participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript. Contributors who do not meet the criteria for authorship should be listed in an acknowledgements section. Examples of those who might be acknowledged include a person who provided purely technical help, writing assistance, or a department chair who provided only general support.

Response: The author’s contribution was added.

2. Requesting deposition of data:
Nucleic acid sequences, protein sequences, and atomic coordinates should be deposited in an appropriate database in time for the accession number to be included in the published article. In computational studies where the sequence information is unacceptable for inclusion in databases because of lack of experimental validation, the sequences must be published as an additional file with the article. Where appropriate, authors should adhere to the standards proposed by the Microarray Gene Expression Data Society (http://www.mged.org) and must deposit microarray data in MIAME-compliant format in one of the public repositories, such as ArrayExpress (http://www.ebi.ac.uk/arrayexpress), Gene Expression Omnibus (GEO; http://www.ncbi.nlm.nih.gov/projects/geo/) or the Center for Information Biology Gene Expression Database (CIBEX; http://cibex.nig.ac.jp).

Response: The protein sequences in the Table 1 have been deposited in the UniProt database